

Author's response to reviews

Title: Prospective study of urinary tract infection surveillance after kidney transplantation

Authors:

Roberto Ribera-Sanchez (cerdea321@yahoo.com.mx)
Dolores Delgado-Ochoa (lolgera@hotmail.com)
Rocio Flores-Paz (rflorespaz@yahoo.com.mx)
Elvia García-Jimenez (elviagen@yahoo.com.ms)
Ramon Espinoza-Hernandez (espinohdzt2003@yahoo.com.mx)
Andres Bazan-Burgosa (bazandr@hotmail.com)
Myriam Arriaga-Alba (arriaga_alba@yahoo.com)

Version: 3 **Date:** 11 January 2010

Author's response to reviews:

BMC Infectious Diseases.

Sir John Kerr ,
Assistant Editor.

With the present, we are sending the new version of the paper "Prospective study of urinary tract infection surveillance after kidney transplantation".

This version has been corrected in accordance of reviewer recommendations. We declare that any of the authors involved on this study, have any competing interest.

This work has only scientific interest to know UTI, as on of the most important complications after renal transplantation, in order to improve the medical assistance to patients. The development of this study did not receive any financial aid from pharmaceutical industry or any other commercial association.

We would appreciate your comments, in order to publish our work in BMC infectious diseases.

Yours faithfully:

Dr. Myriam Arriaga Alba.